Author’s response to reviews

Title: Enhanced antitumor efficacy of cisplatin in combination with HemoHIM in tumor-bearing mice.

Authors:

Hae-Ran Park (hrpark@kaeri.re.kr)
Eun-Jin Ju (eunjeen20@hanmail.net)
Sung-Kee Jo (skjo@kaeri.re.kr)
Uhee Jung (uhjung@kaeri.re.kr)
Sung-Ho Kim (shokim@chonnam.ac.kr)
Sung-Tae Yee (sungtae@sunchon.ac.kr)

Version: 4 Date: 13 March 2009

Author’s response to reviews:

Dear the BioMed Central Editorial Production Team

I correct my manuscript as your indication and I have proof-read my manuscript for typographical errors.

And now, I would like to resubmit my revised manuscript, entitled “Enhanced antitumor efficacy of cisplatin in combination with HemoHIM in tumor-bearing mice”.
